Author's response to reviews

Title: Insulin secretion in patients with latent autoimmune diabetes (LADA): half way between type 1 and type 2 diabetes. Action LADA 9

Authors:

Marta Hernandez (martahernandezg@gmail.com)
Angels Mollo (angelsmollo@gmail.com)
Josep Ramon Marsal (joseprmarsal@yahoo.es)
Aureli Esquerda (aesquerda lleida.ics@gen.cat.cat)
Ismail Capel (Capel7@gmail.com)
Alberto de Leiva (aleiva@santpau.cat)
Manel Puig-Domingo (mpuigd@gmail.com)
Paolo Pozzilli (P.Pozzilli@unicampus.it)
Didac Mauricio (didacmauricio@gmail.com)

Version: 3
Date: 14 December 2014

Author's response to reviews: see over
Dear Editor,

Please, find enclosed the new version of the manuscript entitled ‘Insulin secretion in patients with latent autoimmune diabetes (LADA): half way between type 1 and type 2 diabetes. Action LADA 9’.

There were no additional questions from the reviewers in the link contained in the letter of the editor. There is an additional question from Reviewer 1 concerning his affiliation, but he did not make any additional comments concerning the paper.

Following the Editor’s request we have included in the new version of the manuscript the affiliation of the local Ethics Committees of the two hospitals where patients were recruited.

Sincerely,

Didac Mauricio,